Author’s response to reviews

Title: Perceived stress, reasons for and sources of stress among medical students at Rabigh Medical College, King Abdulaziz University, Jeddah, Saudi Arabia

Authors:
Zohair Gazzaz (zohairgazzaz@hotmail.com)
Mukhtiar Baig (drmukhtiarbaig@yahoo.com)
Bader Al Alhendi (mr.b.alhindi@hotmail.com)
Mohammed Al Suliman (m.alyami220@hotmail.com)
Awshaemah Al Alhendi (Dr.osm1410@hotmail.com)
Mana Al-Grad (m.s.n1993@homail.com)
Mohammed Qurayshah (M_q707@hotmail.com)

Version: 2 Date: 25 Jan 2018

Author’s response to reviews:

Reply to comments

1. In the section ‘Ethics approval and consent to participate’ thank you for clarifying that participant consent was taken in verbal form. As consent was taken in verbal form, please provide the reason it was not provided in written form. If consent was assumed by completion of a survey, please state so.

Reply:
A sentence has been added to page no. 16, line no 387 and 389.

“A verbal consent was taken and moreover, filling out the questionnaire and submission by the student himself was considered as a declaration of willingness to participate. “

2. In the Funding section, please indicate the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Reply:
A sentence has been added to page no. 15, line no 371-373.
“Authors declare that there was no role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.”

3. Please change the heading from ‘Methodology’ to ‘Methods.’

Reply:

The methodology has been changed to methods. (plz check page no 6 line no 144)

4. Upon uploading the revised manuscript, please upload a single, clean version of the manuscript where tracked changes are turned off.

Reply:

Done

5. Please also remove any cover letters or response letters from the file inventory, or documents, which are not intended for publication.

Reply:

Done

6. English language editing

Reply

English language editing has already been done by professional editing services. A certificate of language editing is attached.